Author's response to reviews

**Title:** Early detection and intervention using neutrophil gelatinase-associated lipocalin (NGAL) may improve renal outcome of acute contrast media induced nephropathy: A randomized controlled trial in patients undergoing intra-arterial angiography (ANTI-CIN Study)

**Authors:**

Gernot Schilcher (gernot.schilcher@medunigraz.at)
Werner Ribitsch (werner.ribitsch@medunigraz.at)
Ronald Otto (ronald.otto@klinikum-graz.at)
Rupert H Portugaller (rupert.portugaller@medunigraz.at)
Franz Quehenberger (franz.quehenberger@medunigraz.at)
Martini Truschnig-Wilders (martini.truschnig@klinikum-graz.at)
Robert Zweiker (robert.zweiker@medunigraz.at)
Philipp Stiegler (philipp.stiegler@medunigraz.at)
Marianne Brodmann (marianne.brodmann@medunigraz.at)
Klemens Weinhandl (klemens.weinhandl@web.de)
Joerg H Horina (joerg.horina@medunigraz.at)

**Version:** 3  **Date:** 11 August 2011

**Author's response to reviews:** see over
Dear Christna Chap,

August 11\textsuperscript{th}, 2011

Please excuse our overlooking that an additional change concerning the authors list of our manuscript \textit{“Early detection and intervention using neutrophil gelatinase-associated lipocalin (NGAL) may improve renal outcome of acute contrast media induced nephropathy: A randomized controlled trial in patients undergoing intra-arterial angiography (ANTI-CIN Study)”} in the online submission system had to be made.

MS ID: 2940227405251582

The following correction (major revision) has been made / the minor revision, due to lack of expertise, has not been made. The authors would be very pleased if the BMC Team could undertake the task of correcting the figure format.

\textbf{Major revision:}

\textbf{Author list:} the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs) 'Martini Truschnig' in the submission system, 'Martini Truschnig-Wilders' in the manuscript.

Corrections – Online submission system Author list:
Martini Truschnig-Wilders has been corrected in the online submission system (the correction is underlined); it was not necessary to amend our manuscript.

\textbf{Minor revision:}

We did not any formatting changes to the figure. As offered in your email, we would be very pleased if the formatting specialists of BMC Nephrology could make the requested changes to crop the figure (to minimise white space around the image).
As required by the instructions for authors I confirm that this manuscript has not been published previously and that none of the authors has a conflict of interest related to the work presented in this manuscript. The manuscript has been reviewed by a medical editor with native English.

With best thanks in advance for your attention,

Sincerely,

Joerg H. Horina, M.D.
Associate Professor of Medicine, Specialist in Internal Medicine and Nephrology
Department of Internal Medicine/Division of Nephrology and Hemodialysis
Medical University of Graz
Auenbruggerplatz 27
8036 Graz, Austria
Email: joerg.horina@medunigraz.at
tel: +43 316 385 - 12170
fax: +43 316 385 – 14420